RESEARCH ARTICLE

Population Density, Climate Variables and Poverty Synergistically Structure Spatial Risk in Urban Malaria in India

Mauricio Santos-Vega¹, Menno J Bouma^{2,3}, Vijay Kohli⁴, Mercedes Pascual¹*

1 Department of Ecology and Evolution, University of Chicago, Chicago, United States of America, 2 London School of Hygiene and Tropical Medicine, University of London, United Kingdom, 3 Institute for Climate Sciences (IC3), University of Barcelona, Barcelona, Spain, 4 Ahmedabad Municipal Corporation, Ahmedabad, India

* pascualmm@uchicago.edu

Abstract

Background

The world is rapidly becoming urban with the global population living in cities projected to double by 2050. This increase in urbanization poses new challenges for the spread and control of communicable diseases such as malaria. In particular, urban environments create highly heterogeneous socio-economic and environmental conditions that can affect the transmission of vector-borne diseases dependent on human water storage and waste water management. Interestingly India, as opposed to Africa, harbors a mosquito vector, *Anopheles stephensi*, which thrives in the man-made environments of cities and acts as the vector for both *Plasmodium vivax* and *Plasmodium falciparum*, making the malaria problem a truly urban phenomenon. Here we address the role and determinants of within-city spatial heterogeneity in the incidence patterns of vivax malaria, and then draw comparisons with results for falciparum malaria.

Methodology/principal findings

Statistical analyses and a phenomenological transmission model are applied to an extensive spatio-temporal dataset on cases of *Plasmodium vivax* in the city of Ahmedabad (Gujarat, India) that spans 12 years monthly at the level of wards. A spatial pattern in malaria incidence is described that is largely stationary in time for this parasite. Malaria risk is then shown to be associated with socioeconomic indicators and environmental parameters, temperature and humidity. In a more dynamical perspective, an Inhomogeneous Markov Chain Model is used to predict vivax malaria risk. Models that account for climate factors, socioeconomic level and population size show the highest predictive skill. A comparison to the transmission dynamics of falciparum malaria reinforces the conclusion that the spatio-temporal patterns of risk are strongly driven by extrinsic factors.

Conclusion/significance

Climate forcing and socio-economic heterogeneity act synergistically at local scales on the population dynamics of urban malaria in this city. The stationarity of malaria risk patterns



OPEN ACCESS

Citation: Santos-Vega M, Bouma MJ, Kohli V, Pascual M (2016) Population Density, Climate Variables and Poverty Synergistically Structure Spatial Risk in Urban Malaria in India. PLoS Negl Trop Dis 10(12): e0005155. doi:10.1371/journal. pntd.0005155

Editor: Mathieu Maheu-Giroux, Imperial College London, UNITED KINGDOM

Received: April 12, 2016

Accepted: November 2, 2016

Published: December 1, 2016

Copyright: © 2016 Santos-Vega et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The malaria data was provided by the Ahmedabad Municipal Corporation, and is publicly available from the web site ahmedabadcity.gov.in.

Funding: The authors received no specific funding for this work.

Competing Interests: The authors have declared that no competing interests exist.

provides a basis for more targeted intervention, such as vector control, based on transmission 'hotspots'. This is especially relevant for *P. vivax*, a more resilient parasite than *P. falciparum*, due to its ability to relapse and the operational shortcomings of delivering a "radical cure".

Author Summary

Urbanization and environmental change are the main driving forces of ecological and social change around the globe, specifically in developing countries and for human health. Cities in developing countries exhibit rapid and unplanned urbanization which creates heterogeneous environmental and socio-economic conditions, which can in turn lead to different risks of infection. Here we address the role of urban spatial heterogeneity in infection risk by *Plasmodium vivax* with an extensive surveillance data set from Ahmedabad, India, spanning 12 years. This parasite is one of the four malaria species in humans, and is today the dominant cause of malaria in Indian cities. Our results show clear spatial structure in malaria incidence within the city, dependent on wealth, population density and the climate (temperature and humidity). Because this pattern of spatial risk is largely stationary in time, it can be incorporated in a prediction model and it identifies target 'hotspots' for intervention. Similar conclusions apply to the reported cases of *Plasmodium falciparum* which reinforces the importance of spatial heterogeneity in urban malaria more generally.

Introduction

Addressing health problems associated with urban growth will be one of the major challenges of the 21st century, especially for the developing world [1]. City life is associated with significant variation in socioeconomic and environmental conditions of potential relevance to vector-borne diseases [2–4]. In particular, the pronounced and on-going increase in urban population [5, 6], combined with climate change and economic disparities could act synergistically on the transmission dynamics of malaria [7–9]. Although, Indian cities harbor both malaria parasites, Plasmodium falciparum and Plasmodium vivax, there is an increasing appreciation of the latter as a threat to global health [10], in particular in urban environments where Plasmodium vivax has become the most prevalent parasite. Plasmodium vivax has re-emerged in areas previously cleared of malaria, and has done so with higher mortality and morbidity than previously documented [11-13]. In Indian cities with seasonal transmission, the incidence of vivax malaria starts to rise earlier than that of falciparum malaria, and also earlier than transmission via the vector would allow. This earlier part of the vivax season is dominated and enabled by relapses. The parasite has the ability to delay the development of a fraction of the infectious load of sporozoites in the liver [14], which results in the relapse of the disease after the primary infection is cleared from the bloodstream [15]. The later part of the season is shared with *P. falciparum*, reflecting largely vector-borne transmission following the monsoon rains [10]. A better understanding of the spatial heterogeneity of malaria vivax risk (defined based on the wards' incidence) within cities remains an open area of research in the population dynamics of the disease, of relevance to both prediction and intervention.

Historically, urbanization has led to economic and social transformations associated with profound improvements in sanitation and hygiene [16, 17]. For malaria, the process of

urbanization is generally thought to reduce transmission, primarily because urban environments are largely unsuitable for malaria vectors due to a lack of breeding sites and the pollution of potential larval habitats [18]. Other explanations for reduced malaria risk include better access to health care services and an increased ratio of humans to mosquitoes [2]. There is concern however that areas with rapid unplanned urbanization and poor sanitation may not experience this marked decrease in malaria transmission [5].

The most common hypotheses for the persistence of malaria in cities include spatial variation in: 1) environmental conditions (relative humidity, temperature, precipitation), land use, and stored water, which create a favorable environment for Anopheles breeding in cities [7, 19, 20]; 2) socioeconomic factors (income, human movement, population density and the failure of local malaria intervention among others), which hamper the effectiveness of case management and the promotion of intermittent antimalarials [8, 21–24]; and 3) at more local scales, variation in mosquito behavior and ecology which can influence transmission intensity [22, 23]. Despite increased interest in the role of spatial heterogeneities in the population dynamics of vectorborne diseases [25–28], malaria models typically assume spatially homogeneous transmission and tend to aggregate temporal dynamics over space. In particular, they do not take into account how spatial variation in environmental, climatic and socio-economic conditions affect vector habitat, contact rates, host susceptibility and the effectiveness of control [27–29].

Importantly, these considerations are focused on Africa where endemic malaria remains a predominantly rural problem, because the main mosquito vectors are themselves rural, and in cities, largely peri-urban [2, 3, 5, 7, 8]. By contrast, the Indian subcontinent harbors a truly urban vector, *Anopheles stephensi*, particularly thrives in man-made environments, and breeds in various artificial containers within homes and in water collected in construction sites (catch basins, seepage canals, wells) [30], whereas its sister (sub) species (myorensis) is associated with rural areas. The existence of this particular vector within cities poses a unique challenge to the elimination of malaria in India. Cities can act as a reservoir for the persistence of the disease beyond their administrative limits, and prevent elimination despite considerable gains in the fight against rural transmission. But even beyond the Indian subcontinent, malaria can no longer be considered only a rural problem, given the increasing proportion of the world population living in cities and transmission of the disease in urban environments sometimes surpassing that of rural ones [2, 3, 9].

Here, we describe the spatial pattern of urban *Plasmodium vivax* risk within a large city of India, and investigate its association with socio-economic and environmental heterogeneity. By combining statistical analyses and adapting a probabilistic modeling framework previously proposed for cholera [31], we show that spatial heterogeneity in population size, environmental and economic factors, modulates malaria risk, and that the temporal effect of climate variables on malaria risk interacts with this spatial heterogeneity. These findings emphasize the importance of considering the interplay of climate forcing and socio-economic heterogeneity in the population dynamics of urban malaria in India. They also provide a basis for more targeted intervention, such as vector control, based on identifying transmission 'hotspots'. Comparison to findings for *Plasmodium falciparum* reinforce the evidence for a role of spatial heterogeneity in the transmission of urban malaria in this region.

Methods

Data

We take advantage of a highly disaggregated dataset of monthly malaria cases collected by the Municipal Corporation of the city of Ahmedabad, the capital of state of Gujarat in Northwest India (Fig 1A and 1B). In this largely semi-arid state where malaria is seasonally epidemic,

Ahmedabad reports more than 1000 cases every year. The city presents ideal conditions to investigate malaria transmission dynamics in an urban environment, since it has experienced rapid development, unplanned urbanization and large population growth. It is also located on the banks of the Sabarmati River which creates environmental variation in this otherwise arid setting. The malaria data consist of monthly cases for the dominant parasite, *Plasmodium vivax*, over the last decade, confirmed through microscopy examination of blood slides from clinical (febrile) individuals self-presenting at the health facilities (Fig 1B and 1C). The resulting time series span a total of 12 years (from 2002 to 2014). Because some administrative units (known as wards) were subdivided into smaller wards in 2007, we aggregated the 64 units of the city into 59 wards, in order to maintain the same geographical subdivision through time. For comparison purposes, a similar data set for *Plasmodium falciparum* is also considered.

Apart from decadal census data, annual population data were provided by the Ahmedabad Municipal Corporation to approximate the population of each ward. Socio-economic data were obtained from the District Census Handbook of the concerned district for the year 2001 from the Directorate of Census Operations, Gujarat. Monthly time series (from 2002 to 2014) for mean temperature, mean rainfall and relative humidity at 8 am are those from the meteorological station of the city of Ahmedabad (and were provided by the Indian Institute of Technology in Gandhinagar).



Fig 1. Study Area. Location of study area (A), and temporal patterns of incidence of *P. falciparum* (red solid) and *P. vivax* (blue dashed) (B, C). Boxplots are shown in B to illustrate seasonality, and time series are shown in C to illustrate interannual variation.

doi:10.1371/journal.pntd.0005155.g001

Spatial regularities in malaria transmission

In order to investigate the existence of a spatial pattern in malaria incidence within the city of Ahmedabad, we performed a series of statistical analyses to address whether malaria risk varied within the city and what factors explained this variation. First, we analyzed the spatial and temporal variation of malaria incidence and identified regions of high and low risk based on incidence. Second, we performed a series of statistical analyses on the role of socioeconomic and environmental factors in the spatial, and spatio-temporal patterns of malaria incidence. These analyses ranged from a simple t-test comparing socio-economic factors between the two regions of differential malaria risk, to time series models incorporating the autocorrelation in the data and the external drivers (including climatic ones), to a full spatio-temporal general linear mixed model with random effects. Third, based on these results a probabilistic dynamical model was formulated for malaria transmission at the ward level, and predictions of this model were evaluated at the city level.

To consider a measure of vivax malaria risk independent from interannual variation, we normalized malaria incidence for each ward in a given year by the total number of cases throughout the city. We then ranked these normalized values across wards to determine if high risk locations were consistently so over time. To examine the robustness of the patterns, we complemented the estimation of the intensity of infection with the Slide Positivity Rate (SPR) [32] S1 Fig 1. SPR, defined as the number of laboratory-confirmed malaria cases per 100 suspected cases examined, provides a rapid and inexpensive means of assessing the burden of malaria in the population that relies on health care facilities. SPR provides a better estimate of the true intensity of malaria than malaria incidence (reported cases divided by population), as it is more robust to spatial sampling errors/biases resulting from variation in the population size of the wards [32]. SPR is independent from incidence, although it can be prone to different biases (as the result for example of other febrile diseases which would increase the number of blood tests taken). Confirming that the patterns are consistent across these two measures of malaria intensity is important because the population denominator that is relevant for the reporting may vary in ways that are difficult to determine. Also, the local population size might be under- or overestimated by a census carried out only once every 10 years in India.

To further characterize spatial variation in risk we applied a 2k-means cluster algorithm to the incidence data at the wards level, and examined the existence of at least two regions differing in malaria risk. (We pre-determined two groups of wards to consider the hypothesis of different transmission intensity in the core and periphery of the city). Also, two groups allow us to consider the subdivision of wards into high and low risk regions. We hypothesized that differences in malaria risk in these two main areas are largely explained by demographic and socio-economic factors. To test if the two regions differed significantly in those variables, we first extracted socioeconomic indicators, including slum density (number of slums/ward area), unemployment (number of unemployed people), number of marginal workers, literacy, population below 6 years, total population, number of households, vulnerable and economically deprived communities, from the 2001 Ahmedabad census and calculated the density of slums per ward based on cartographic information on the slums' distribution within the city <u>S2 Fig</u>. A t-test was applied to evaluate if these two regions differed significantly in these covariates.

We then addressed whether the temporal variation in malaria incidence responded differentially to climate variables (rainfall, temperature and humidity) across the two regions. To determine which predictors explain the temporal variation in vivax cases, we considered models with autoregressive terms to account for serial correlation in the data. The correlation structure in the malaria time series was assessed by inspecting the autocorrelation function ACF. Then we applied a generalized linear model (GLM) framework. Because observed count data, such as reported cases in infectious diseases, often exhibit significant over-dispersion [33], a negative binomial distribution of cases was used. Backwards model selection was based on the minimum Akaike Information Criterion (AIC) (S1 Table). Variables with coefficient significantly different from zero were selected. Spatial correlation was incorporated by assuming a conditional autoregressive (CAR) process in the random effect v_i :

$$v_i v_j = N\left(rac{\sum_j a_{ij} v_j}{\sum_j a_{ij}}, rac{\sigma_v^2}{\sum_j a_{ij}}
ight)$$

where σ^2 controls the strength of the local dependence, and a_{ij} are neighborhood weights for each ward based on distance to the river. We additionally compared the best model to a model with a different distribution (Zero Inflated Poisson) but this model was not significantly better (S2 Table). Finally, we considered the full spatio-temporal variation at the level of wards, by fitting a generalized linear mixed model including as covariates the effects of temperature and humidity, and a spatially-structured random effect weighted by the distance to the river (SSRE). Parameters and their distributions were estimated with Bayesian Markov Chain Monte Carlo (MCMC) parameter sampling implementation in WINBUGS.

Probabilistic model for malaria dynamics

To model malaria risk within the city of Ahmedabad an inhomogeneous Markov chain model was used, following the theoretical framework developed for cholera by Reiner et al. 2012. In this approach the monthly malaria cases are categorized into discrete states of malaria incidence, which we chose as "low malaria", "mild malaria," and "high malaria". The three discrete states partitioned the distribution of monthly incidence based on the 25th, lower than 75th and above 75th quantiles. Then, the model assigns baseline probabilities $P_{i,j}$ to the transitions between these states in a defined time step as described by the following transition probability matrix P:

$$P_{ij} = \begin{bmatrix} P_{1,1} & (1 - P_{1,1} - P_{1,3}) & P_{1,3} \\ P_{2,1} & (1 - P_{2,1} - P_{2,3}) & P_{2,3} \\ P_{3,1} & (1 - P_{3,1} - P_{3,3}) & P_{3,3} \end{bmatrix}$$

Where the baseline probabilities depend on the state of the system only in the previous time step. Modification of this basic matrix can introduce the effect of covariates, including the consideration of different regions such as the two groups of wards, identified in previous analyses. Thus, the resulting Markov chain model can be made inhomogeneous by allowing transition probabilities to depend on temporal and spatial environmental drivers; here seasonality, the state of the neighboring wards, temperature, Relative humidity and socioeconomic heterogeneity (summarized by the two groups of wards):

$$P_{i,j,k,t} = P_{i,j,d} \cdot Seas_{i,j,t,d} \cdot Neigh_{i,j,v,d} \cdot temp_{t,d} \cdot RH_{t,d}$$

Where $P_{i,j,k,t}$ is the probability that ward k goes from state i to j from time t to time t+1. This probability is dependent on: (1) $P_{i,j,d}$, the baseline transition probabilities of moving from state i to state j for a ward in risk region d; (2) a seasonal factor $Seas_{i,j,t,d} = (1 + \beta_{i,j,d})^{Se_{(t,d)}}$, where the seasonality exponent $Se_{(t,d)}$ is periodic over the 12 months of the year and each group d has its own seasonality; (3) a neighborhood effect $Neigh_{i,j,v,d} = (1 + \alpha_{i,j,d})_{t,d}^v$, where v corresponds to the malaria state (0,1,2) of the neighboring ward with the highest value. This function reduces to 1 when none of the neighboring wards have malaria reported. The parameters of these

functions, $\beta_{i,j,d}$ and $\propto_{i,j,d}$ are estimated. Finally, the effects of temperature and humidity are included as sigmoidal functions, similar to the formulation for ENSO and its effect on cholera in Reiner et al 2012: $temp/Rh_{i,j,t,d} = \left(A_d tan\left(h_d \frac{Temp/RH_{t-\gamma}}{(2M_d)}\right)tan\left(\frac{h_d}{2}\right)\right)$ where $Temp/Rh_{t,i,j,d}$ is the temperature or humidity with temporal lag γ , in this case 2 months for the humidity and 1 month for the temperature, A is the maximal amplitude, M controls the scale (normalization by the maximum values allowed for the climate factors), and h is shape factor varying between 0 and ~ π to go from a linear to a nonlinear effect (S3 Fig).

We considered different models obtained by including or neglecting the effect of a subset of the following factors: temperature, relative humidity, the state of the neighboring wards and the two different risk regions. We compared each of the models to a null model) employing a likelihood ratio test. The most complex model has 78 parameters (S3 Table). Under Markovian assumptions, the transitions for the different time steps (months) are independent from each other [31, 34], which allows us to explicitly write a likelihood. The constraint that each transition probability must be between 0 and 1 was imposed by a barrier method (i.e. by setting the probability to 0 whenever its estimated value falls outside these limits) [31]. Each model was fitted by maximizing the likelihood with a Nelder-Meade simplex algorithm, which allows for the incorporation of such constraints.

Finally, to assess prediction performance, a cross-validation approach was implemented by sequentially removing the epidemic months (August-November) that follow the monsoon in a given year, refitting the model to the remaining data, and simulating it four months ahead starting from August to predict the course of the seasonal outbreak for the omitted period. Forecasting accuracy was estimated by computing the likelihood of the observed state. To that end, we inferred the probability distribution of the predicted state by performing 5000 independent simulations. This procedure is then sequentially repeated removing, one at a time, all the epidemic seasons available. To quantify the accuracy of our predictions, we calculate the percentage mean absolute error in our predictions, as well as a second quantity more practical and possibly relevant to public health, based on the definition of a 'large' outbreak. We defined such as event as one where the peak of the epidemic at the whole city level exceeds the 75% quantile of the distribution of this quantity. We quantify the fraction of times the model correctly predicts the observed malaria incidence state (above the 75% quantile). Then, to examine and illustrate the importance of the climate covariates to the predictions, we simulated the model using different combinations of humidity and temperature 'data'. In particular, predictions for the epidemic months in the anomalous, low incidence years, were obtained using: 1) monthly observed temperature and average humidity 2) monthly average temperature and observed humidity 3) monthly average humidity and average temperature. Monthly averages were computed based on the mean of all previous years for a given month. We also examined the performance of the model by obtaining a one-step ahead prediction, where we removed 1 month of data at a time for all the wards (S4 Fig).

Results

We initially addressed whether the spatial distribution of normalized *Plasmodium vivax* risk is heterogeneous throughout the city, and how the ranking of risk varies in time. The top left panel of Fig 2 shows that several locations systematically rank high or low based on their *P. vivax* incidence through time. The top right panel in this figure reveals a stable spatial regularity of the locations with the highest malaria burden through time. This remarkable temporal stability of the spatial pattern suggests the existence of strong underlying determinants that are largely stationary at the temporal scales of malaria transmission within the city.



rig 2. Analysis of spatio-temporal patterns of malaria vivax incidence in the city of Anmedabad. The panels show the distribution of the cases normalized by population, with the intensity of the color (from low yellows to high reds) corresponding to the ranking of incidence. There is striking consistency from one year to the next in the places exhibiting the highest burden of the disease. Some of this regularity also extends to the two parasites. See S2 Fig, for comparison with the patterns obtained with SPR (Slide Positivity Rate) as an alternative measure of malaria intensity see S1 Fig.

doi:10.1371/journal.pntd.0005155.g002

The spatial pattern observed in Fig 2 also indicates the existence of two distinct malaria risk regions within Ahmedabad: one comprised of the wards close to the Sabarmati River and the core of the city (region 1), and the other, of those in the newer urban periphery (region 2). Fig 3 shows the results of the cluster analysis using the rankings data, demonstrating that the group of wards that are close to the river and in the inner part of the city (referred to as high risk region hereafter) have a different malaria risk than those in the in the periphery (low risk region). *Plasmodium vivax* cases in the region defined as high risk are significantly (p<0.001) higher than those in the region defined as low risk, whereas the seasonal pattern remains the same for both areas. This results hold for both parasites S5 Fig.

We then asked whether the differences in malaria risk between the low and high malaria risk regions are associated with differences in population density (number of slums, population size, number of households) or economic level (income, unemployment, literacy). Table 1 summarizes the results from statistical comparisons (t-test) between the high and low risk regions (for *P. vivax*) based on socioeconomic indicators from the 2001 census. We found that the high risk region has significantly higher unemployment, slum density, total population and number of households. In addition, we did not find significant differences in the mean area of the wards. The differences in slum density, unemployment, literacy, economically disadvantaged people and vulnerable communities (both Scheduled Castes, SC, and Scheduled Tribes, ST, the official designations given to two different groups of historically disadvantaged



significant differences in the annual malaria incidence between the two regions (p<0.001), for both parasites.

doi:10.1371/journal.pntd.0005155.g003

people) are pronounced, with the high risk region encompassing for example a density of slums that is at least 1.2 higher than that of the low risk region (Table 1).

Moreover, the temporal variation in malaria incidence between the two regions could be influenced differentially by the environmental covariates. Table 2 show the results of the stepwise multiple linear regression between *Plasmodium vivax* cases and climate covariates (temperature, humidity and rainfall) and S6 Fig show the residuals of our best model for each region. The temporal variation in the low risk region responds predominantly to changes in temperature and humidity, whereas that of the high risk region is explained predominantly by humidity, although both parameters are seasonally associated with rainfall. Humidity would act in both the temporal and spatial dimensions, as the water table and associated ground

Table 1. Statistical analysis of differences between the two regions for P. vivax, based on the socioeconomic information of the 2001 census.

variable statistic.t		parameter.df	p.value	log(mean of High risk)	log(mean of low risk)	
Slum density	1.834	54.99	0.042	3.261	2.747	
Unemployment	3.059	29.21	0.004	10.82	8.828	
Marginal workers	3.197	30.56	0.003	6.967	5.592	
Literacy	3.217	29.18	0.003	10.884	8.566	
pop_below_6years	3.275	29.43	0.002	9.095	7.120	
Total population	3.237	29.18	0.003	11.20	8.808	
Area	0.336	45.51	0.737	1.379	1.438	
Number of Households	3.178	29.29	0.003	9.565	7.544	
Vulnerable communities	3.210	40.75	0.002	6.207	4.825	
Economically deprive communities	3.934	36.14	0.000	8.841	6.682	

doi:10.1371/journal.pntd.0005155.t001

Low risk region						
	Estimate	Std. Error	z value	Pr(> z)	2.50%	97.50%
ar1	0.63891	0.064	9.917	0	0.448	0.718
intercept	-1.257	2.126	-0.591	0.554	-13.702	9.159
temp	0.140	0.069	2.006	0.044	-0.035	0.725
RH	0.034	0.025	1.329	0.0183	0.027	0.258
High risk region						
	Estimate	Std. Error	z value	Pr(> z)	2.50%	97.50%
ar1	0.5835	0.0689	8.4637	0.0000	0.4483	0.7186
intercept	-2.2716	5.8323	-0.3895	0.6969	-13.7027	9.1594
RH	0.1775	0.0235	3.2948	0.0010	0.1332	0.3952

Table 2. Results of the best model for monthly cases as a function of environmental covariates for Plasmodium vivax.

doi:10.1371/journal.pntd.0005155.t002

.

moisture should be higher in the high risk region closer to the river, and this parameter is known to affect survival of the adult mosquitoes [35]. Thus, increased humidity (through changes in the water table) can lead to an earlier onset of the *Plasmodium vivax* season and to higher incidence <u>S7 Fig</u>. Finally, results for the general linear mixed model show that the best model includes the effects of temperature and humidity, and the random spatial variation in the data weighted by the distance to the river (<u>S4 Table</u> and <u>S5 Table</u>).

Interestingly, *Plasmodium falciparum* also exhibits stable regularities in risk levels within the city <u>S8 Fig</u>. We find the existence of two different regions with contrasting incidence, largely consistent with those for *Plasmodium vivax* Fig 3, and a significant difference in socio-economic level and environmental conditions for these two regions (<u>S6 Table</u> and <u>S7 Table</u>).

For a more dynamical perspective, we used an inhomogeneous Markov chain model that incorporates the effect of spatial and temporal variation on malaria risk. Results are also consistent for both parasites. Specifically, the comparison of the different models analyzed (S3 Table and S8 Table), shows that the best model is the one that accounts for the effects of seasonality, neighbors, temperature, humidity, and includes the two different risk regions identified above (model 5 Table 3). To test the significance of the individual components of each model (alternative hypothesis) against a model including seasonality only (model number 1, null hypothesis) we employed a likelihood ratio test. Improvements in likelihood for models 3 to 8 are significant, and so are the effects of seasonality, temperature, neighbors and the two regions (p<0.05). Improvements in likelihood for model 2 (seasonal effect + neighborhood effect) are not significant at p = 0.05. Moreover, models that account for the effect of spatial heterogeneity, represented by including the two risk regions, tend to perform better than those that do not incorporate this effect (S3 Table and S8 Table).

Table 3. Likelihood comparison of the models showing the covariates included in each model. The best likelihood is for the model that incorporates seasonality, temperature, two regions and neighbors. The last column shows the result of a likelihood ratio test between the null model (model 1) and each of the other models.

models	seasonality	Neighbors	Temperature	RH	Regions	Log lik	DF	AIC	LRT
Model 5	+	+	+	+	+	-4821.559	78	9799.119	*
Model 6	+	+	+		+	-4903.554	72	9951.107	*
Model 7	+	+			+	-5670.935	66	11473.87	*
Model 4	+	+	+	+		-5894.961	39	11867.92	*
Model 8	+				+	-5932.35	54	11972.7	*
Model 3	+	+	+			-6573.846	36	13219.69	*
Model 2	+	+				-6648.966	33	13363.93	
Model 1	+					-6656.621	27	13367.24	-
							· · · · · · · · · · · · · · · · · · ·		·

doi:10.1371/journal.pntd.0005155.t003





Fig 4. Comparison of observed and predicted cases with the best model. In (A), the red line corresponds to the average number of cases per 1000 for the 59 wards. The blue dots correspond to predictions given by the median of 5000 simulations, and the gray bars correspond to the 5th and 95th percentiles. In (B-D), simulations of the model predict the seasonal epidemics of 2009 and 2010 starting from the end of the monsoons (August) under modifications of the observed climate covariates. The different panels show the effect of fixing temperature and/or humidity at their mean monthly values, to remove their effect on the interannual variation of these anomalous years. When the interannual effect of both is removed (B), the model clearly overestimates the cases. Individual effects are less pronounced (C, D) although predictions are also higher than observations. Our best model has a mean absolute error of 68% for predicting the peak of the epidemic in a year with a high number of cases (2013). Fig 4 (E, left), shows the distribution of model forecasts from 5000 runs for October 2003 based on October 2002 data. Although the mean prediction differs from the observation, almost all (~84%) model simulations resulted in large events for 2003. The figure on the right repeats this hindcast analysis for October 2013 (using data from 2012). Here, we find a reduced but still large (~87%) probability of a large outbreak.

doi:10.1371/journal.pntd.0005155.g004

Finally, Fig 4 shows the predictions for the epidemic months for *Plasmodium vivax*, generated with the cross-validation procedure described above and the best model, which includes seasonality, temperature, the state of the neighboring wards, and spatial heterogeneity. The predicted rate of cases for the peak of the season is coherent with the observed rate. Further explorations of the model show the importance of temperature and humidity in the prediction of malaria risk Fig 4B. For example, the pronounced dip in the cases for years 2009 and 2010 is captured by the dependencies of the model on temperature and humidity, as those two years are dryer, warmer and less humid Fig 4C and 4D. Although our best model exhibits a tendency to under-predict the size of the peaks which we discuss below, it is able to capture the seasonality and to a reasonable extent, the interannual variation in the data. The anomalous decrease in the number of cases in 2009 and 2010 seasons can be explained by the variation in the climate covariates. Because the model is stochastic and it considers discrete states (no malaria, low and high), we simulated repeatedly, and from these ensemble of simulations computed the mean number of cases in a given month for a given ward. Our simulations generate realizations of the stochastic process and therefore, configurations of the discrete states. To convert the discrete states to cases, we used the mean number of cases for each class. The red line corresponds to the wards mean observed cases of the city. The blue dots show the median of the simulated values and the blue shaded regions correspond to the 5th– 95th percentile range over 5000 simulations.

Discussion

Most transmission models of vector-borne diseases tend to aggregate the data at large scales and treat transmission homogeneously in space [27–29]. However, at local scales spatial heterogeneity can significantly influence the risk of infection. In particular, urban environments can exhibit pronounced heterogeneity from rapid and unplanned urbanization. Spatial heterogeneity in the environmental conditions such as temperature and humidity or in socioeconomic level can affect mosquito ecology, such as habitat distribution, vector longevity, biting rate or host finding ability [36], and in factors related to human exposure and susceptibility respectively [8, 16, 37].

For *Plasmodium vivax* in Ahmedabad, we found defined heterogeneity in malaria risk that is slow-changing and therefore largely 'stationary' in time, relative to the characteristic temporal scales of the population dynamics of the disease. The presence of such stable pattern suggests strong and spatially-structured determinants of malaria risk. In particular, the existence of two main regions with different risk was shown to be associated with socioeconomic level. Higher risk is largely concentrated in the inner part of the city where socio-economic indicators reveal higher poverty on average (Fig 3 and Table 1). These results are consistent with the previously described negative associations between malaria and socioeconomic status [38–41]. Disease persistence decreases with increasing employment, literacy and income S1 Fig 9, with poor people more vulnerable to ineffective diagnosis and treatment for financial and cultural reasons, and less able to access antimalarial and anti-mosquito protection [42]. Consistent results for *Plasmodium falciparum* emphasize the driven nature of the patterns.

The two different risk regions within the city were also shown to exhibit differential temporal responses to climate forcing. This finding underscores the importance of humidity to malaria transmission, with a higher water table in the high risk region possibly increasing relative humidity and affecting vector ecology. The spread of malaria requires favorable conditions for the survival of both the mosquito and the parasite. Temperatures in the approximate range of 21°-32°C and a relative humidity of at least 60% are most conducive to transmission [38]. Malaria vectors need to live at least 8 days in order to transmit malaria, and higher humidity increases both survival rate and activity rate [43, 44]. These relationships explain why Anopheles stephensi is more active and prefers feeding during the night when relative humidity is higher. The low risk region should have lower humidity (ground moisture) given its distance from the Sabarmati river and other water bodies. It has been established that if the average monthly relative humidity (measured at 8 am) is below 60%, then the lifespan of the mosquito is too short leading to very low or no malaria transmission [4, 45, 46]. The effect of humidity is also evident in the estimated regression coefficients (Table 2), where the value corresponding to humidity is one order of magnitude higher for the high than the low risk region. This means that a typical annual change of 10% in humidity with the rest of the covariates kept the same for both regions, will result in an increase of 403 to 1030 additional malaria cases (or 15% to 39% relative to the mean) in the high-risk region, and 70 to 411 additional cases (or 7% to 33%

relative to the mean) in the low risk region. Rainfall itself, which is closely related to both humidity and temperature, is not retained in the regression as a significant explanatory variable probably because of its collinear effects with the other climatic factors. Our results from the mixed model indicate however that while climate factors (temperature and humidity) play an important role in disease transmission, it is their combined effect with spatial heterogeneity in the risk weighted by the distance to the Sabarmati river that better explains the spatio-temporal variation in the malaria incidence. More mechanistic models informed by time series data on disease incidence and vector abundance, will provide further insight on the role and interplay of climate covariates through their effects on parasite mortality, as well as on vector breeding and longevity.

Social and economic elements such as the quality of housing can also favor the biological development of mosquitoes [47]. It is common in urban areas of India for water to be supplied irregularly; this leads to water storage within houses which creates multiple breeding sites for the mosquito in overhead tanks, cisterns and cement tanks [48]. Higher population density would result in higher water storage concentrations in close proximity to people. At the local scale spatial heterogeneity in urban malaria risk would follow from high density, especially where coexisting with poverty, as well as environmental microclimates from the proximity of water bodies.

Our dynamical and stochastic model captures the seasonal pattern and the main trends in the interannual variation of the malaria cases. Interestingly, most of the models that incorporate spatial structure, namely the two regions, perform better than the models that do not. This conclusion is consistent with the results for diarrheal diseases in Dhaka, where consideration of different parts of the city also improved model performance [31, 48]. Our best model identifies significant spatial effects at two different scales: (1) that of neighboring wards (p<0.01), where the probability of transitioning to a higher risk level depends of the level of the surrounding wards, and (2) that of the two regions (p<0.001) influenced socioeconomic and demographic level. Although our model under-predicts the size of outbreaks, this tendency is expected from the discretization of the cases into a small number of levels, which preserves the rankings of the cases over time but tends to reduce the magnitude of the peaks. Predictions in that scale can still be useful when evaluating the risk of an outbreak larger than a given selected threshold [31]. Although additional classes could be incorporated in the formulation of the model, this would rapidly increase the number of parameters.

The model is able to capture the interannual trends and in particular, the lower outbreaks of years like 2009 and 2010, based on the effect of climate covariates. These two years exhibit anomalous high temperature and low humidity (associated with low monsoon rainfall), only comparable to values in 2002, another year with low incidence S10 Fig. The predictions of Fig 4B, for temperature and humidity kept at monthly averages, show that the model over-predicts the number of cases Fig 4. Thus, the anomalous incidence is explained by the lower temperature and higher humidity. Fig 4C and 4D isolate the effect of each climate covariate on the prediction, showing a stronger effect of temperature on the temporal reduction of cases of those two years.

Besides prediction, the phenomenological modeling framework applied here is also useful to address the spatial scale at which to aggregate the data to consider process-based epidemio-logical models, in a way that balances reducing the noise with representing dominant spatial heterogeneity. Questions on the spatial scale of aggregation are specifically relevant to address-ing climate forcing in the context of socio-economic heterogeneity. Given that pronounced changes in urbanization will co-occur with those in climate, these are fundamental questions for infectious disease dynamics within cities.

Although our approach is able to capture the interannual variation in the data and predict the peak of the epidemic, it could be improved in several directions. For example, one could incorporate in the model: (1) mobility fluxes derived from the spatial distribution of the

population with movement models, to replace the near-neighbor effects on transition probabilities; (2) the explicit effect of population density on group-dependent parameters explicitly; (3) further analysis of the local effect of environmental heterogeneities such as river discharge and soil moisture on malaria incidence at higher resolution by increasing the number of groups in the model. Moreover, temporal changes of the city itself would be of interest, including changes in the local speed of urbanization, and their implications for mobility, population distribution and economic level.

The region has experienced strong malaria interventions in the last three decades reflected in the pronounced negative trends in the number of reported cases from the 1980s and 19990s to the 2000s. From 2000 onwards, malaria prevalence in the city of Ahmedabad has remained however fairly stationary. Although the spatio-temporal variation in intervention efforts could influence the results of our models. This is unlikely given that the interventions within the city are largely homogeneous in space (S11 Fig), with small differences between the high and low risk regions.

Our probabilistic model tends to underpredict the size of outbreaks. This bias results from the transformation of the incidence data into discrete malaria levels, which smooth's out extreme events. The effect of this bias can be assessed and corrected by lowering the threshold probability (the proportion of simulations with large outbreaks) below 50% using ROC (Receiver-Operating-Curves) (Reiner et al., 2012). The number of discrete classes describing the malaria levels could also be increased or estimated to balance complexity and accuracy. At the limit, one could move to stochastic models that do not require such discretization, although their parameterization would present challenges related to model complexity.

A better understanding of urbanization and malaria is needed, since urban environments can contribute to the persistence of the disease and frustrate elimination efforts more broadly at a regional level, by creating a reservoir for the disease in cities that contributes to transmission in rural areas. In India, the earlier National Eradication Programs focused on rural areas, with urban malaria contributing to the resurgence of disease in the 1970s [48]. Urbanization and growing populations also exacerbate inequalities in access to water, and in so doing introduce variation in another fundamental but poorly understood environmental factor for vector-transmitted infections. In concert these two dimensions, environmental and socio-economic, define the relevant spatial scales at which to address transmission dynamics. They also define the existence of spatial 'hotspots' of high disease risk in urban environments, especially those of the developing world where economic inequality and variation in disease vulnerability can be pronounced. Identifying these hotspots for targeted intervention in urban environments, can contribute to the control of vector-transmitted diseases, and eventually to the elimination of malaria in India.

Supporting Information

S1 Fig. Analysis of spatio-temporal patterns of malaria slide positivity rate in the city of Ahmedabad. The panels show the distribution of SPR with the intensity of the color (from low yellows to high reds for *P. vivax* and from light yellow to blue for *P. falciparum*) corresponding to the ranking based on the intensity of the transmission. (TIF)

S2 Fig. Slum distribution in the city (left) and slum density calculation (right). The latter was generated by overlaying the slum distribution map with the wards map provided by the municipal corporation, and calculating the number of slums per ward divided by the ward area.

(TIF)

S3 Fig. Functional forms of the humidity (bottom) and temperature (top) effects illustrating the flexibility of the formulation, with different values of parameter h leading to different shapes. (Here we show these functional forms for an arbitrary amplitude A and scale M, and different shape values: h = 0 (black line), h = 7/8pi (red line) and h = pi (green line). (TIF)

S4 Fig. Comparison of predictions and observations. The red line corresponds to the average monthly cases per 10000 for the 59 wards. The blue dotted line corresponds to one-month ahead predictions for the median of the 5000 simulations values, and the light brown shaded region corresponds to the interval between the 5th and 95th percentiles for these simulations. (TIF)

S5 Fig. Malaria seasonal pattern for the two risk regions. The top panels represent the seasonal pattern for *Plasmodium falciparum*, for the high risk region in the left and the low risk region in the right. The bottom panel shows the corresponding patterns for *Plasmodium vivax*. (TIF)

S6 Fig. The figure shows a time plot and ACF of the residuals for the fitted models. Although most of the autocorrelations fall within the confidence intervals, there is a small autocorrelation at lags of 11 and 12 months (seen in the significant spike of the ACF plot). This suggests that the model can be slightly improved by capturing the remaining seasonal variation.

(TIF)

S7 Fig. The top panels show scatterplots of humidity vs cases for *P. vivax (left panels)* and *P. falciparum (right panels)*, and the bottom ones, the time series for the cases (in red for vivax and in blue for falciparum) together with those for humidity (in black). (TIF)

S8 Fig. Analysis of spatio-temporal patterns of malaria incidence in the city of Ahmedabad (from low yellow to high blue). The panels show the distribution of the cases normalized by population, with the intensity of the color corresponding to the ranking of incidence. (TIF)

S9 Fig. Scatterplot of socioeconomic indicators (number of households, literacy and unemployment) and the number of cases of *P. falciparum* and *P. vivax* per ward. (TIF)

S10 Fig. From left to right, scatterplots of the annual mean temperature, humidity and rainfall vs annual *Plasmodium vivax* incidence.

S11 Fig. The map depicts the mean number of containers treated with larvicide throughout the city. Red dots represent wards in the high risk regions, and yellow dots, those in the low risk region.

(TIF)

S1 Table. Comparisons of nested models based on the likelihood ratio test (where asterisks indicate significance relative to the previous model). (DOCX)

S2 Table. The table shows comparisons of the zero inflated Poisson and negative binomial models fitted to the data. (DOCX)

S3 Table. Different parameterizations of the probabilistic model. The table shows the effects included in the model formulation and the corresponding total number of parameters. (DOCX)

S4 Table. Model comparisons highlight the best model which incorporates the random effects, as well as the effect of temperature and humidity. (DOCX)

S5 Table. Estimated parameters for the best model which includes the effects of temperature relative and relative humidity, and the random effects, whose values are significantly different from zero. Confidence intervals (CI) from posterior distributions (from two chains that are well mixed and have converged). (DOCX)

S6 Table. Statistical analysis of differences between the two regions identified using the socioeconomic information of the 2001 census for *P. falciparum*. (DOCX)

S7 Table. Results of the best negative binomial model selected by AIC for *Plasmodium falciparum* with climate covariates. This model includes an autoregressive term that accounts for serial autocorrelation, the effect of relative humidity in both regions, and the effect of temperature in the low risk region. (DOCX)

S8 Table. Likelihood comparison of the models showing the covariates included in each model. The best likelihood is obtained for the model that incorporates seasonality, temperature, two regions and the effect of neighbors. The last column shows the likelihood ratio test between the null model (model 1) and each of the other models. (DOCX)

Author Contributions

Conceptualization: MSV MP MJB.

Data curation: MSV MJB VK.

Formal analysis: MSV.

Investigation: VK.

Methodology: MSV MP.

Software: MSV.

Validation: MSV.

Writing - original draft: MSV MP MJB.

Writing - review & editing: MSV MP MJB.

References

- 1. Harpham T. 2009 Urban health in developing countries: what do we know and where do we go? Health Place 15, 107–116. doi: 10.1016/j.healthplace.2008.03.004 PMID: 18455952
- Hay S. I., Guerra C. A., Tatem A. J., Atkinson P. M., & Snow R. W. (2005). Urbanization, malaria transmission and disease burden in Africa. Nature Reviews. Microbiology, 3(1), 81–90. doi: 10.1038/ nrmicro1069 PMID: 15608702

- 3. Omumbo J. A., Hay S. I., Snow R. W., Tatem A. J. & Rogers D. J. 2005 Modelling malaria risk in East Africa at high-spatial resolution. Trop. Med. Int. Heal. 10, 557–566.
- 4. Dye C. 2008 Health and urban living. Science 319, 766–769. doi: 10.1126/science.1150198 PMID: 18258905
- Keiser J., Utzinger J., Caldas De Castro M., Smith T. A., Tanner M. & Singer B. H. 2004 Urbanization in sub-Saharan Africa and implication for malaria control. In American Journal of Tropical Medicine and Hygiene, pp. 118–127. PMID: 15331827
- 6. United Nations. World Urbanization Prospects: The 2014 Revision, Highlights (ST/ESA/SER.A/352) New York: United Nations DoEaSA, Population Division; 2014.
- Robert V., Macintyre K., Keating J., Trape J. F., Duchemin J. B., Warren M. & Beier J. C. 2003 Malaria transmission in urban sub-Saharan Africa. Am. J. Trop. Med. Hyg. 68, 169–176. PMID: 12641407
- Donnelly M. J. et al. 2005 Malaria and urbanization in sub-Saharan Africa. Malar. J. 4, 12. doi: 10.1186/ 1475-2875-4-12 PMID: 15720713
- Reiner R. C., Smith D. L. & Gething P. W. 2015 Climate change, urbanization and disease: summer in the city... Trans. R. Soc. Trop. Med. Hyg. 109, 171–172. doi: 10.1093/trstmh/tru194 PMID: 25491136
- Roy M, Bouma MJ, Ionides EL, Dhiman RC, Pascual M. The Potential Elimination of Plasmodium vivax Malaria by Relapse Treatment: Insights from a Transmission Model and Surveillance Data from NW India. PLoS Negl Trop Dis. 2013;7.
- Baird JK. Neglect of Plasmodium vivax malaria. Trends in Parasitology. 2007. pp. 533–539. doi: 10. 1016/j.pt.2007.08.011 PMID: 17933585
- Sattabongkot J, Tsuboi T, Zollner GE, Sirichaisinthop J, Cui L. Plasmodium vivax transmission: Chances for control? Trends in Parasitology. 2004. pp. 192–198. doi: 10.1016/j.pt.2004.02.001 PMID: 15099559
- 13. Price RN, Tjitra E, Guerra CA, Yeung S, White NJ, Anstey NM. Vivax malaria: Neglected and not benign. Am J Trop Med Hyg. 2007; 77: 79–87. PMID: 18165478
- 14. White N (2011) Determinants of relapse periodicity in *Plasmodium vivax* malaria. Malaria J 10: 297.
- 15. Warrall DA (1993) Clinical features of malaria. In: Guiles HM, Warrell DA, Bruce-Chwatt LJ (eds), Bruce-Chwatt's essential malariology, Arnold, London, p35–49.
- Gross R., Schell B., Molina M. C., Leão M. A. & Strack U. 1989 The impact of improvement of water supply and sanitation facilities on diarrhea and intestinal parasites: a Brazilian experience with children in two low-income urban communities. Rev. Saude Publica 23, 214–220. PMID: 2617116
- Klinkenberg E., McCall P., Wilson M. D., Amerasinghe F. P. & Donnelly M. J. 2008 Impact of urban agriculture on malaria vectors in Accra, Ghana. Malar. J. 7, 151. doi: <u>10.1186/1475-2875-7-151</u> PMID: <u>18680565</u>
- Qi Q., Guerra C. a, Moyes C. L., Elyazar I. R. F., Gething P. W., Hay S. I. & Tatem A. J. 2012 The effects of urbanization on global Plasmodium vivax malaria transmission. Malar. J. 11, 403. doi: <u>10.1186/1475-</u> 2875-11-403 PMID: 23217010
- Trape J. F., Lefebvre-Zante E., Legros F., Ndiaye G., Bouganali H., Druilhe P. & Salem G. 1992 Vector density gradients and the epidemiology of urban malaria in Dakar, Senegal. Am. J. Trop. Med. Hyg. 47, 181–189. PMID: 1354414
- Afrane Y. A., Klinkenberg E., Drechsel P., Owusu-Daaku K., Garms R. & Kruppa T. 2004 Does irrigated urban agriculture influence the transmission of malaria in the city of Kumasi, Ghana? Acta Trop. 89, 125–134. PMID: 14732235
- Noor A. M., Zurovac D., Hay S. I., Ochola S. A. & Snow R. W. 2003 Defining equity in physical access to clinical services using geographical information systems as part of malaria planning and monitoring in Kenya. Trop. Med. Int. Heal. 8, 917–926.
- Hay S. I. et al. 2009 A world malaria map: plasmodium falciparum endemicity in 2007. PLoS Med. 6, 0286–0302.
- Stoddard S. T., Morrison A. C., Vazquez-Prokopec G. M., Soldan V. P., Kochel T. J., Kitron U., Elder J. P. & Scott T. W. 2009 The role of human movement in the transmission of vector-borne pathogens. PLoS Negl. Trop. Dis. 3.
- 24. De Silva P. M. & Marshall J. M. 2012 Factors contributing to urban malaria transmission in sub-saharan Africa: A systematic review. J. Trop. Med.
- 25. Hasibeder G. & Dye C. 1988 Population dynamics of mosquito-borne disease: persistence in a completely heterogeneous environment. Theor. Popul. Biol. 33, 31–53. PMID: 2897726
- Cosner C., Beier J. C., Cantrell R. S., Impoinvil D., Kapitanski L., Potts M. D., Troyo A. & Ruan S. 2009 The effects of human movement on the persistence of vector-borne diseases. J. Theor. Biol. 258, 550– 560. doi: 10.1016/j.jtbi.2009.02.016 PMID: 19265711

- Reiner R. C. et al. 2013 A systematic review of mathematical models of mosquito-borne pathogen transmission: 1970–2010. J. R. Soc. Interface 10, 20120921. doi: 10.1098/rsif.2012.0921 PMID: 23407571
- Acevedo M. A., Prosper O., Lopiano K., Ruktanonchai N., Caughlin T. T., Martcheva M., Osenberg C. W. & Smith D. L. 2015 Spatial Heterogeneity, Host Movement and Mosquito-Borne Disease Transmission. PLoS One 10, e0127552. doi: 10.1371/journal.pone.0127552 PMID: 26030769
- Perkins T. A., Scott T. W., Le Menach A., & Smith D. L. (2013). Heterogeneity, Mixing, and the Spatial Scales of Mosquito-Borne Pathogen Transmission. PLoS Computational Biology, 9(12).
- Cator L. J., Thomas S., Paaijmans K. P., Ravishankaran S., Justin J. a, Mathai M. T., Read A. F., Thomas M. B. & Eapen A. 2013 Characterizing microclimate in urban malaria transmission settings: a case study from Chennai, India. Malar. J. 12, 84. doi: 10.1186/1475-2875-12-84 PMID: 23452620
- Reiner R. C., King A. A., Emch M., Yunus M., Faruque A. S. G. & Pascual M. 2012 Highly localized sensitivity to climate forcing drives endemic cholera in a megacity. Proc. Natl. Acad. Sci. 109, 2033–2036. doi: 10.1073/pnas.1108438109 PMID: 22308325
- Bouma M. J., Dye C., van der Kaay H. J. 1996 Falciparum malaria and climate change in the northwest frontier province of Pakistan. Am J Trop Med Hyg. 55(2):131–7. PMID: 8780449
- 33. Lawless J. F. 1987 Negative binomial and mixed poisson regression. Can. J. Stat. 15, 209–225.
- Finger F., Knox A., Bertuzzo E., Mari L., Bompangue D., Gatto M., Rodriguez-Iturbe I. & Rinaldo A. 2014 Cholera in the Lake Kivu region (DRC): Integrating remote sensing and spatially explicit epidemiological modeling. Water Resour. Res. 50, 5624–5637.
- Metha D.R. 1934 Studies on the longevity of some Indian anophelines. Part I. Survival of Anopheles subpictus Grassi under controlled conditions of temperature and humidity. Rev. app. Ent. B 23 (1) 18– 19.
- Smith D. L., Dushoff J. & McKenzie F. E. 2004 The risk of a mosquito-borne infection in a heterogeneous environment. PLoS Biol. 2, e368. doi: 10.1371/journal.pbio.0020368 PMID: 15510228
- Bansal S., Grenfell B. T., & Meyers L. A. (2007). When individual behaviour matters: homogeneous and network models in epidemiology. Journal of the Royal Society, Interface / the Royal Society, 4(16), 879–891.
- Bui H. M., Clements A. C. A., Nguyen Q. T., Nguyen M. H., Le X. H., Hay S. I., ... Horby P. (2011). Social and environmental determinants of malaria in space and time in Viet Nam. International Journal for Parasitology, 41(1), 109–16. doi: 10.1016/j.ijpara.2010.08.005 PMID: 20833173
- Sachs J., & Malaney P. (2002). The economic and social burden of malaria. Nature, 415(6872), 680– 685. doi: 10.1038/415680a PMID: 11832956
- Worrall E., Basu S. & Hanson K. 2005 Is malaria a disease of poverty? A review of the literature. Trop. Med. Int. Heal. 10, 1047–1059.
- **41.** Levin S. a. (2007). Introduction: Infectious diseases. Environment and Development Economics, 12 (05), 625–626.
- Ajayi I. O., Jegede A. S., Falade C. O. & Sommerfeld J. 2013 Assessing resources for implementing a community directed intervention (CDI) strategy in delivering multiple health interventions in urban poor communities in Southwestern Nigeria: a qualitative study. Infect. Dis. Poverty 2, 25. doi: <u>10.1186/2049-</u> 9957-2-25 PMID: 24156481
- 43. Snodgrass R. E. 1939 The Principles of Insect Physiology. Science (80-.). 90, 159–159.
- Gaaboub I., El-Sawaf S., El-Latif M. 1971Effect of different relative humidities and temperatures on eggproduction and longevity of adults of Anopheles (Myzomyia) pharoensis Theob. Zeitschrift Für Angewandte Entomologie. 67(1–4):88–94
- 45. Kessler S. & Guerin P. M. 2008 Responses of Anopheles gambiae, *Anopheles stephensi*, Aedes aegypti, and Culex pipiens mosquitoes (Diptera: Culicidae) to cool and humid refugium conditions. J. Vector Ecol. 33, 145–149. PMID: <u>18697317</u>
- **46.** Mayne B. 1930 A study of the influence of relative humidity on the life and infectibility of the mosquito. Indian J Med Res. 17(4):1119–1137.
- Kirby, Matthew J., et al. "Risk factors for house-entry by malaria vectors in a rural town and satellite villages in The Gambia." Malaria Journal 7.1 (2008): 2.
- 48. Sharma V. P. 1996 Re-emergence of malaria in India. Indian J. Med. Res. 103, 26–45. PMID: 8926025